Author’s response to reviews

Title: Anti-angiogenesis therapy based on the bone marrow-derived stromal cells genetically engineered to express sFlt-1 in mouse tumor model

Authors:

min hu (gym_w@163.com)
jinlinag yang (jlyang01@163.com)
teng hong (hong_teng@163.com)
yongqian jia (Jia-Yq@Yahoo.com)
rui wang (raiwang@sohu.com)
xiaowei zhang (sean53@163.com)
yang wu (stillwy@163.com)
yan luo (yanluo300@sina.com)
xiancheng chen (xcchen@sohu.com)
ru zhang (zhangru237@sohu.com)
ling tian (tl09168@hotmail.com)
xia zhao (xzhao@sina.com)
yuquan wei (yuquawei@hotmail.com)

Version: 5 Date: 25 September 2008

Author’s response to reviews:

Dear editor,

Based upon the Editorial Production Team’s advices, several major and minor revisions have been made in the final version.

These are major revisions:

The list of authors in the manuscript is now exactly as they are in the submission system.

Superscript numbers of references were changed.

Reference 4 was substituted by another paper of the same research team.

Figures were rearranged to six files.

Figure legends were revised in accordance with figures.

The minor revisions in the key words, references, figure legends figure size were made one by one accordingly.

With best wishes

Minhu